Author's response to reviews

Title: Elevated MED28 expression predicts poor outcome in women with breast cancer

Authors:

Nam K Yoon (namkeunyoon@gmail.com)
Erin L Maresh (emaresh@mednet.ucla.edu)
Yahya Elshimali (elshimali@gmail.com)
Ai Li (liaipk@gmail.com)
Steve Horvath (shorvath@mednet.ucla.edu)
David B Seligson (dseligson@mednet.ucla.edu)
David Chia (dchia@mednet.ucla.edu)
Lee Goodglick (lgoodglick@mednet.ucla.edu)

Version: 7 Date: 18 June 2010

Author's response to reviews: see over
Dear Editorial Production Team:

We have made all of the modifications that you mentioned. Specifically:

1. We now state that all authors read and approved the final manuscript (page 13).
2. All figures are not on only one page (i.e., single page composite). One of the Figures before, was on 2 pages (Figure 2). We have now made this 2 separate figures. Therefore, there are now a total of 5 Figures in the paper (instead of 4). These modifications have been made in the body of the paper.
3. In the text, we now refer to the Supplementary material as "additional files":
4. We have removed all headers from the Text, Figures, and Additional Files.
5. We are not quite sure what your request means to include the Additional files section "on a portrait page". We have resubmitted this material, however, with Figures now on separate PowerPoint Files

If you have any questions or concerns, please don't hesitate to contact me.

Sincerely,

Lee Goodglick

Associate Professor,
Director, UCLA EDRN
Department of Pathology and Laboratory Medicine
UCLA School of Medicine
Center for the Health Sciences
10833 Le Conte Ave.
Los Angeles, CA 90095